Author’s response to reviews

Title: Comparison of expectations and beliefs about good teaching in an academic day release medical education program

Authors:

Thea ACM van Roermund (dvanroermund@hotmail.com)
Henk G Mokkink (Henk.Mokkink@radboudumc.nl)
Ben JAM Bottema (Ben.Bottema@radboudumc.nl)
Chris van Weel (Chris.vanWeel@radboudumc.nl)
Albert JJA Scherpbier (A.scherpbier@maastrichtuniversity.nl)

Version: 3  Date: 27 August 2014

Author’s response to reviews: see over
Dear Mr Ulep,

Thank you for accepting our manuscript for publication in BMC Medical Education.

We inserted the revision, as requested, in the method section as follows:
“The research method used complies with the Qualitative research review guidelines – RATS."

On behalf of all co-authors I would like to submit our revised manuscript.

Sincerely,
Thea van Roermund